Author's response to reviews

Title: Edge restenosis: Impact of low dose irradiation on cell proliferation and ICAM-1 expression

Authors:

Rainer Voisard (rainer.voisard@uni-ulm.de)
Jochen Hoeb (jochen.hoeb@uni-ulm.de)
Regine Baur (regine.baur@uni-ulm.de)
Tina Herter (tina.herter@uni-ulm.de)
Andreas Hannekum (andreas.hannekum@uni-ulm.de)
Vinzenz Hombach (vinzenz.hombach@uni-ulm.de)

Version: 6 Date: 6 July 2006

Author’s response to reviews:

Ulm, Germany, July 6th, 2006

Dear BMC Editorial Team,

enclosed please find revision II of our manuscript entitled: “Edge restenosis: Impact of low dose irradiation on cell proliferation and ICAM-1 expression”.

The authors would like to thank the referees for fair and constructive criticism. The queries of the referees have been carefully addressed.

With best wishes

Priv.-Doz. Dr. Rainer Voisard